Author’s response to reviews

Title: The antifibrotic drug pirfenidone inhibits spondyloarthritis fibroblast-like synoviocytes and osteoblasts in vitro

Authors:

Julie Stougaard (jkl@biomed.au.dk)
Søren Lomholt (soren.lomholt@gmail.com)
Pernille Ommen (pernille.ommen.andersen@post.au.dk)
Jens Kelsen (jenskels@rm.dk)
Tue Kragstrup (kragstrup@biomed.au.dk)

Version: 8 Date: 18 Sep 2018

Author’s response to reviews:

Editor Comments:

1. It was found that a few crucial subsections of the Declarations were missing in your latest revised version of your manuscript. Please ensure that each of the following subsections are included in your Declarations:

   a. Ethics approval and consent to participate
   b. Consent for publication
   c. Availability of data and materials
   d. Competing interests
   e. Funding
   f. Authors’ contributions
   g. Acknowledgements
   h. Authors’ information (optional)

Please see our submission guidelines for further details about what should be included in each section:
https://bmrheumatol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

Please note that any sections that are not applicable to your manuscript should still be included with the statement “Not applicable.”

ANSWER: Thank you. I added the subsections.

2. In your Competing interests subsection of the Declarations, please be sure to declare that author Tue Wenzel Kragstrup is an Editorial Board Member for BMC Rheumatology.

ANSWER: I added this information.

3. It was noticed in your Authors’ contributions subsection that you described the contributions of a “JK” and a “JKS.” While there is an individual in the author list who corresponds with JK, there does not seem to be one who corresponds with JKS. If this is a typographical error and the initials are meant to represent the same author, please take this opportunity to consolidate the contributions under the initials JK.

ANSWER: JK is Jens Kelsen. JKS is Julie Stougaard, I changed it to JS.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.